Author’s response to reviews

Title: Prevalence and risk factors for chronic kidney disease of unknown cause in Malawi: a cross-sectional analysis in a rural and urban population

Authors:

SOPHIE HAMILTON (s.hamilton16@imperial.ac.uk)
Wisdom P Nakanga (wisdom.nakanga@kpsllmw.org)
Josephine E Prynn (Josephine.Prynn1@lshtm.ac.uk)
Amelia C Crampin (Mia.Crampin@lshtm.ac.uk)
Daniela Fecht (d.fecht@imperial.ac.uk)
Paolo Vineis (p.vineis@imperial.ac.uk)
Ben Caplin (b.caplin@ucl.ac.uk)
Neil Pearce (Neil.Pearce@LSHTM.ac.uk)
Moffat J Nyirenda (Moffat.Nyirenda@LSHTM.ac.uk)

Version: 3 Date: 15 Aug 2020

Author’s response to reviews:

Dear Ms Lowe,

Thank you very much for your review of our manuscript:

BNEP-D-19-01054R2
Prevalence and risk factors for chronic kidney disease of unknown cause in Malawi: a cross-sectional analysis in a rural and urban population

We have noted the following changes and upload a clean and re-ordered version of the manuscript:

1. List of Abbreviations
   – We note that some abbreviations used in the manuscript text and figure legends/tables, such as "MK," have not been included in the ‘List of Abbreviations’ section. Please ensure that all abbreviations present in the text are listed in this section.
   Response: Thank you for picking this up, this has been amended with all abbreviations included

2. Formatting
Please move the 'Declarations' section to just after the list of abbreviations in the main manuscript, before the References.

Please include all tables and figures at the end of the manuscript, after the References section.

Response: these have been changed and both main tables and figures have been added to the end of the manuscript

3. Acknowledgements
–Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

Response: this has been changed

4. Clean Manuscript
–At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: the clean version has been uploaded

We thank you for your review of our manuscript and are very happy to make any further suggestions required

With best wishes,

Sophie Hamilton